Author's response to reviews

Title: Rapidly progressing, fatal acute promyelocytic leukaemia that initially manifested as a painful third molar: a case report

Authors:

Juan A Suárez-Cuenca (jsuarez@ifc.unam.mx)
José L Arellano-Sánchez (md_arlos79@hotmail.com)
Aldo A Scherling-Ocampo (aldosterona1@hotmail.com)
Gerardo Sánchez-Hernández (gmtp@prodigy.net.mx)
David Pérez-Guevara (dvdprz@hotmail.com)
Juan R Chalapud-Revelo (injuan@hotmail.com)

Version: 6 Date: 19 May 2009

Author's response to reviews: see over
May 18th, 2009.

Dr. Sadath
JMCR Editorial Team

Thank you for your commitment.

We appreciate the comments from reviewers about the manuscript, 1209737221229598 – Rapidly progressing fatal acute promyelocytic leukaemia first manifested as painful third molar: a case report.

Accordingly, we have revised the affiliations section, according to the format - Department, Institute, (Street), City, (Postal code), Country. Please find the changes in red.

We appreciate your kind consideration of our manuscript. Please let us know of any additional requirements.

Best Regards,

Juan Antonio Suárez-Cuenca, MD, PhD.
Internal Medicine, Hospital General Ticomán, Xoco, SSGDF.
Cellular Physiology Institute, National University of Mexico.
México D.F., MEXICO.
Rapidly progressing, fatal acute promyelocytic leukaemia that initially manifested as a painful third molar: a case report

Juan Antonio Suárez-Cuenca\textsuperscript{1,2,3*}, José Luis Arellano-Sánchez\textsuperscript{1,2}, Aldo Alfonso Scherling-Ocampo\textsuperscript{1,2}, Gerardo Sánchez-Hernández\textsuperscript{2}, David Pérez-Guevara\textsuperscript{4} and Juan Ramón Chalapud-Revelo\textsuperscript{5}

**Addresses:** [S1]
\textsuperscript{1} Internal Medicine Department, Ticomán and \textsuperscript{2} Xoco General Hospitals, SSDF. Av. México Coyoacán s/n, esq. Bruno Traven, Mexico City, P.O. 30340, Mexico.
\textsuperscript{3} Cellular Biology Department (Postgraduate Program in Biomedical Science), Institute of Cellular Physiology, Circuito Exterior s/n Ciudad Universitaria, Mexico City, P.O. 04510, Mexico.
\textsuperscript{4} Maxillofacial Surgery Department (Division of Research and Postgraduate Studies), Faculty of Dentistry, Av. Institutos s/n Ciudad Universitaria, Mexico City, P.O. 04510, Mexico.
\textsuperscript{5} Haematology Department, National Cancer Institute, Av. San Fernando No. 22, Mexico City, P.O. 14080, Mexico.

**Email addresses**
JASC: jsuarez@ifc.unam.mx
JLAS: md_arlos79@hotmail.com
AASO: aldosterona1@hotmail.com
GSH: gmtp@prodigy.net.mx
DPG: dvdprz@hotmail.com
JRCR: injuan@hotmail.com

*Corresponding author*
Juan Antonio Suárez-Cuenca
Phone: (005255) 5622-5666. Fax: (005255) 5622-5611.
e-mail: jsuarez@ifc.unam.mx